Author’s response to reviews

Title: Associations between BMI and 10 candidate genes at ages 4, 7 and 10 in a large UK sample

Authors:

Claire MA Haworth (Claire.Haworth@iop.kcl.ac.uk)
Lee M Butcher (Lee.Butcher@iop.kcl.ac.uk)
Sophia J Docherty (Sophia.Docherty@iop.kcl.ac.uk)
Jane Wardle (J.Wardle@ucl.ac.uk)
Robert Plomin (Robert.Plimin@iop.kcl.ac.uk)

Version: 2 Date: 2 August 2007

Author’s response to reviews: see over
Dear Dr Norton,

Re: Revised Manuscript: ‘Associations between BMI and 10 Candidate Genes at Ages 4, 7 and 10 in a Large UK Sample’. (MS No. 9388663821535587).

Please find attached our revised manuscript entitled Associations between BMI and 10 Candidate Genes at Ages 4, 7 and 10 in a Large UK Sample for your consideration for publication in BMC Medical Genetics.

As requested, we have included in the methods section a statement concerning the ethical approval and informed consent for our study.

Contact details for the corresponding author are as follows:
Claire Haworth, SGDP Centre, Box P080, Institute of Psychiatry, De Crespigny Park, London SE5 8AF. Tel: +44 (0) 207 - 848 5401, email: claire.haworth@iop.kcl.ac.uk.

I look forward to hearing from you.

Sincerely,

Claire Haworth